Author’s response to reviews

Title: Ultrasound Detects Synovitis in Replaced and Other Surgically Operated Joints in Rheumatoid Arthritis Patients

Authors:

Evan Choate (EChoate@mednet.ucla.edu)
Gurjit Kaeley (Gurjit.Kaeley@jax.ufl.edu)
Jenny Brook (JBrook@mednet.ucla.edu)
Roy Altman (journals@royaltman.com)
John Fitzgerald (JFitzgerald@mednet.ucla.edu)
Astrid Floegel-Shetty (AFloegelShetty@mednet.ucla.edu)
David Elashoff (DElashoff@mednet.ucla.edu)
Veena Ranganath (vranganath@mednet.ucla.edu)

Version: 3 Date: 15 Nov 2019

Author’s response to reviews:

Dear Assistant Editor,

Thank you for your letter informing us that our manuscript is potentially acceptable for publication in BMC Rheumatology, once we have carried out some essential revisions.

1. Trial Registration Number- This was included in the last section of the abstract.
2. Funding- We described the role of the funding bodies.
3. Clean manuscript- We have submitted one clean version without tracked changes.

Thank you again. Please let us know if there are any further questions or comments.

Best,
Veena

Veena K Ranganath, MD, MS
Associate Clinical Professor
Division of Rheumatology, UCLA